

Serum potassium and mortality risk in hemodialysis patients: a cohort study

Rooij, E.N.M. de; Dekker, F.W.; Cessie, S. le; Hoorn, E.J.; Fijter, J.W. de; Hoogeveen, E.K.; Netherlands Cooperative Study Adeq

Citation

Rooij, E. N. M. de, Dekker, F. W., Cessie, S. le, Hoorn, E. J., Fijter, J. W. de, & Hoogeveen, E. K. (2022). Serum potassium and mortality risk in hemodialysis patients: a cohort study. *Kidney Medicine*, *4*(1). doi:10.1016/j.xkme.2021.08.013

Version:Publisher's VersionLicense:Creative Commons CC BY 4.0 licenseDownloaded from:https://hdl.handle.net/1887/3277731

Note: To cite this publication please use the final published version (if applicable).

Serum Potassium and Mortality Risk in Hemodialysis Patients: A Cohort Study

Esther N.M. de Rooij, Friedo W. Dekker, Saskia Le Cessie, Ewout J. Hoorn, Johan W. de Fijter, and Ellen K. Hoogeveen; on behalf of the Netherlands Cooperative Study on the Adequacy of Dialysis-2 (NECOSAD) Study Group

Rationale & Objective: Both hypo- and hyperkalemia can cause fatal cardiac arrhythmias. Although predialysis serum potassium level is a known modifiable risk factor for death in patients receiving hemodialysis, especially for hypokalemia, this risk may be underestimated. Therefore, we investigated the relationship between predialysis serum potassium level and death in incident hemodialysis patients and whether there is an optimum level.

Study Design: Prospective multicenter cohort study.

Setting & Participants: 1,117 incident hemodialysis patients (aged >18 years) from the Netherlands Cooperative Study on the Adequacy of Dialysis-2 study were included and followed from their first hemodialysis treatment until death, transplantation, switch to peritoneal dialysis, or a maximum of 10 years.

Exposure: Predialysis serum potassium levels were obtained every 6 months and divided into 6 categories: \leq 4.0 mmol/L, >4.0 mmol/L to \leq 4.5 mmol/L, >4.5 mmol/L to \leq 5.0 mmol/L, >5.0 mmol/ L to \leq 5.5 mmol/L (reference), >5.5 mmol/L to \leq 6.0 mmol/L, and >6.0 mmol/L.

Outcomes: 6-month all-cause mortality.

Analytical Approach: Cox proportional hazards and restricted cubic spline analyses with time-dependent

Accumulating evidence points to the adverse impact of Adyskalemia on life expectancy in patients receiving hemodialysis.¹ Both hypo- and hyperkalemia can cause potentially fatal cardiac arrhythmias and sudden death.² Potassium homeostasis is mainly regulated by the kidneys, responsible for excreting 90% of the dietary potassium intake. Patients receiving hemodialysis rely mainly on potassium removal during each dialysis session.² Hyperkalemia, defined as a serum potassium level of ≥5.5 mmol/ L, is a common electrolyte disorder occurring in 12%-20% of patients receiving hemodialysis.3-6 Nonadherence to dietary potassium restrictions and metabolic acidosis increase the risk of hyperkalemia. Relatively little attention is paid to hypokalemia and a low potassium level, defined as serum potassium levels of <3.5 mmol/L and <4.0 mmol/ L, respectively, occurring in 2% and 13% of patients receiving hemodialysis, respectively.^{3,5,6} Malnourishment, metabolic alkalosis, potassium binders, and low potassium dialysate are risk factors for hypokalemia.^{1,5} In addition,

predialysis serum potassium levels were used to calculate the adjusted HRs for death.

Results: At baseline, the mean age of the patients was 63 years (standard deviation, 14 years), 58% were men, 26% smoked, 24% had diabetes, 32% had cardiovascular disease, the mean serum potassium level was 5.0 mmol/L (standard deviation, 0.8 mmol/L), 7% had a low subjective global assessment score, and the median residual kidney function was 3.5 mL/min/1.73 m² (IQR, 1.4-4.8 mL/min/1.73 m²). During the 10-year follow-up, 555 (50%) deaths were observed. Multivariable adjusted HRs for death according to the 6 potassium categories were as follows: 1.42 (95% CI, 1.01-1.99), 1.09 (95% CI, 0.82-1.45), 1.21 (95% CI, 0.94-1.56), 1 (reference), 0.95 (95% CI, 0.71-1.28), and 1.32 (95% CI, 0.97-1.81).

Limitations: Shorter intervals between potassium measurements would have allowed for more precise mortality risk estimations.

Conclusions: We found a U-shaped relationship between serum potassium level and death in incident hemodialysis patients. A low predialysis serum potassium level was associated with a 1.4-fold stronger risk of death than the optimal level of approximately 5.1 mmol/L. These results may imply the cautious use of potassium-lowering therapy and a potassium-restricted diet in patients receiving hemodialysis.

Visual Abstract included

Complete author and article information (including a list of the members of the Netherlands Cooperative Study on the Adequacy of Dialysis-2 [NECOSAD] Study Group) provided before references.

Correspondence to E.N.M. de Rooij (e.n.m. rooij@lumc.nl)

Kidney Med. 4(1):100379. Published online October 22, 2021.

doi: 10.1016/ j.xkme.2021.08.013

© 2021 The Authors. Published by Elsevier Inc. on behalf of the National Kidney Foundation, Inc. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

patients receiving hemodialysis have low intracellular potassium concentrations, despite the tendency of hyperkalemia development in them. Total body potassium levels can be up to 10% lower in patients receiving hemodialysis than in controls and is associated with an increased risk of hypertension, cardiovascular disease, and death.^{7,8}

The optimal predialysis serum potassium level is unknown. There are no randomized controlled trials that have examined the target predialysis potassium level with regard to long-term outcomes. Therefore, we have to rely on prospective cohort and registry studies. In the general population, serum potassium levels between 3.5 and 5.0 mmol/L are considered to be within the normal range. In patients with chronic kidney disease, the optimal range is between 4.0 and 4.5 mmol/L.⁹ An important limitation to previous studies investigating the relationship between serum potassium levels and death on hemodialysis is the inclusion of mainly prevalent instead of incident patients, which may have resulted in survivor bias.¹⁰ Furthermore,



Kidney Medicine

PLAIN-LANGUAGE SUMMARY

Both high and low serum potassium levels can cause fatal heart arrhythmias. In patients receiving hemodialysis, the mortality risk of low serum potassium levels, in particular, may have been underestimated. Therefore, we investigated the relationship between predialysis serum potassium level and 6-month mortality in 1,117 patients receiving hemodialysis during 10 years of follow-up. In our analysis, we adjusted for the potential confounders of age, sex, diabetes, cardiovascular disease, smoking, residual kidney function, and nutritional state. We found that low (\leq 4.0 mmol/L) and high (>6.0 mmol/L) predialysis serum potassium levels are associated with a 1.4- and 1.3-fold, respectively, stronger risk of death than the optimal serum potassium level of approximately 5.1 mmol/L. These results may imply the cautious use of potassium-lowering therapy and a potassium-restricted diet in patients receiving hemodialysis.

until now, cohort studies have estimated the risk of low serum potassium levels in comparison with relatively low levels (4.0-4.5 mmol/L) as a reference category and, therefore, most likely underestimated the mortality risk (hazard ratios [HRs], 1.03-1.14) related to hypokalemia.^{4,11,12}

Low predialysis serum potassium level is a potentially modifiable risk factor; however, its adverse effect on death in patients receiving hemodialysis may be underestimated. Therefore, we studied the relationship between serum potassium levels and death in a prospective cohort of incident hemodialysis patients and investigated whether there is an optimum level to pursue. Results from this study may inform future guidelines for patients receiving hemodialysis.

METHODS

Study Design and Population

The Netherlands Cooperative Study on the Adequacy of Dialysis-2 is a prospective multicenter cohort study of patients, aged >18 years, with incident end-stage renal disease, starting with their first dialysis treatment, as previously described in detail.¹³ Briefly, enrollment occurred between 1997 and 2007 at 38 dialysis centers throughout the Netherlands. The maximum follow-up was 10 years after the start of hemodialysis, with the latest follow-up date on January 1, 2018. The institutional review board of the Academic Medical Hospital, Amsterdam, the Netherlands approved the study (approval number, MEC95/226a) and the institutional review boards of all participating hospitals confirmed this by an additional local approval. All patients gave written informed consent. Follow-up visits were scheduled at 3 months after the start

of dialysis therapy, at 6 months, and subsequently at intervals of 6 months. All centers predominantly used a dialysate potassium concentration of 2 mmol/L and, if indicated, 3 mmol/L. Baseline was defined as 3 months after the start of hemodialysis treatment, when the patients' fluid and metabolic conditions had stabilized. Dates of mortality were immediately reported during follow-up and ascertained by a nephrologist.

The cohort comprised 1,117 patients receiving hemodialysis, without previous kidney replacement therapy, 3 months after starting dialysis. Survival time was defined as the number of days between 3 months after the start of the hemodialysis treatment (baseline) and the date of death, the date of censoring due to loss to follow-up, kidney transplantation, transfer to a nonparticipating dialysis center, a switch to peritoneal dialysis treatment, or the end of the follow-up.

Data Collection

Demographic and clinical data such as age, sex, ethnicity, primary kidney disease, current smoking, medication, a history of diabetes, cardiovascular disease, malignancy or chronic lung disease, Subjective Global Assessment (SGA), blood pressure, blood, and 24-hour urine samples were collected at the start of dialysis treatment and at all visits until the end of follow-up. Primary kidney disease was classified according to the codes of the European Renal Association-European Dialysis and Transplant Association.¹³ We grouped patients into 4 classes of primary kidney disease: diabetic nephropathy, glomerulonephritis, renal vascular disease, and other kidney diseases. A history of diabetes was defined based on diabetes mellitus registered as a comorbid condition or diabetic nephropathy as primary kidney disease. Current smoking was defined as current cigarette smokers, including those who quit smoking in the past 3 months. Cardiovascular disease was defined as any history of a cerebral vascular accident, a myocardial infarction, or peripheral vascular disease.

For blood pressure, the mean systolic and diastolic blood pressure values prior to dialysis over the previous 2 weeks were calculated. The nutritional state was measured using a 1-7 score on SGA, with scores of 6-7 indicating normal protein-energy wasting, scores of 4-5 indicating moderate protein-energy wasting, and scores of 1-3 indicating severe protein-energy wasting.¹⁴

Serum potassium, albumin, creatinine, and urea levels were measured during the prelude to dialysis and after the long dialysis interval. In addition, urea and creatinine levels were also measured in the urine. Residual kidney function was estimated using combined urea and creatinine clearance and corrected for body surface (mL/min/1.73 m²).

Statistical Analysis

Variables are presented as means \pm standard deviations (SDs), median (interquartile range), or numbers (proportions) where appropriate and according to 6 predefined baseline predialysis serum potassium level categories: ≤ 4

mmol/L, >4 mmol/L to \leq 4.5 mmol/L, >4.5 mmol/L to \leq 5.0 mmol/L, >5.0 mmol/L to \leq 5.5 mmol/L, >5.5 mmol/L to \leq 6.0 mmol/L, and >6.0 mmol/L.

We assessed the relationship between predialysis serum potassium level and all-cause mortality during 10 years of follow-up using several methods. In all analyses, survival was measured from 3 months (baseline) after the start of hemodialysis. The survival probabilities for the 6 predialysis serum potassium categories at baseline were visualized using Kaplan-Meier curves for the first 10 years of follow-up.

For all following analyses, the predialysis serum potassium level was included as a time-dependent variable and updated at a 6-month interval after the start of the first hemodialysis treatment. First, we used life tables to calculate absolute mortality rates during the follow-up within each of the 6 time-dependent predialysis serum potassium level categories. Second, we used a timedependent Cox proportional hazards model to calculate crude and multivariable adjusted HRs for 6-month allcause mortality during 10 years of follow-up. As normal serum potassium levels vary from 3.5 to 5.5 mmol/L and no optimum level has been definitively recommended within that range, the predialysis serum potassium category (>5.0 to \leq 5.5 mmol/L) with the lowest mortality rate in our cubic spline analysis was considered the reference category.¹⁵ Analyses were adjusted for potential confounders measured at baseline: age, sex, current smoking, history of diabetes, history of cardiovascular disease, residual kidney function, and SGA score (full model). We did not control for time-varying confounding because later values of time-dependent confounders may be influenced by earlier levels of predialysis serum potassium. Therefore, adjustments for such time-dependent variables could have introduced bias.¹⁶ Third, the continuous relationship between time-dependent predialysis serum potassium level and mortality was explored by modeling a 4-knot restricted cubic spline with 95% confidence intervals (CIs) adjusted for the previously mentioned confounders. The knots were chosen at the 5th, 35th, 65th, and 95th percentiles of the predialysis serum potassium level distribution.¹

We performed 4 sensitivity analyses. First, we repeated our full time-dependent Cox proportional hazards model with additional adjustments for serum phosphate level, serum bicarbonate level, serum albumin level, and normalized protein catabolic rate to assess any residual confounding by nutritional state. Second, we performed time-dependent Cox proportional hazard analyses considering as a secondary outcome cardiac death due to dyskalemia, cardiac arrest, myocardial infarction, and sudden death. We did not include cardiac death in our primary analysis as specific causes of death are more likely to be misclassified. Third, we repeated our analyses without censoring for patients switching to peritoneal dialysis. In the primary analyses, we censored patients from the moment they switched from hemodialysis to peritoneal dialysis, since we aimed to investigate only the effect of the predialysis serum potassium level on the mortality risk in patients receiving hemodialysis. However, this might have resulted in informative censoring, since patients switching to peritoneal dialysis are often in better or worse clinical condition than the general population receiving hemodialysis. Fourth, we repeated our Cox proportional hazards model using the baseline predialysis serum potassium level category as a fixed variable to calculate the HRs for 10-year all-cause mortality and cardiac death. This enabled us to evaluate whether the relationship between predialysis serum potassium level and all-cause mortality is indeed a short-term effect that can be better estimated using a timedependent analysis.

We assumed missing values to be missing at random. Missing data were handled using 2 different strategies. For missing predialysis serum potassium levels at baseline (n = 9; 1%), we carried the next observed predialysis serum potassium level backward, and for the missing values during the 10-year follow-up (n = 869; 13%), we carried the last observed predialysis serum potassium level forward. For the following missing baseline data—smoking (n = 8; 1%), history of diabetes (n = 14; 1%) or history of cardiovascular disease (n = 14; 1%), residual kidney function (n = 259; 23%), and SGA score (n = 82; 7%), we used multiple imputations to avoid bias and maintain power, using 10 imputations, and including all relevant baseline variables and the outcome in the model.¹⁸

In the proportional hazards regression models, the proportionality assumption for each covariate was checked by adding a product term between that covariate and the logarithm of follow-up time. All analyses were performed using SPSS 23.0 (International Business Machines Corporation) and R version 3.5.1 (R Core Team).

RESULTS

Baseline Characteristics

Of all Netherlands Cooperative Study on the Adequacy of Dialysis-2 study participants, 1,117 patients receiving hemodialysis survived the first 3 months after starting dialysis and were therefore included in the analysis. At baseline, the mean age of the study cohort was 63 years (SD, 14 years), 58% were men, 26% were current smokers, 24% had a history of diabetes, 32% had a history of cardiovascular disease, 7% had a low SGA score, and the median residual kidney function was 3.5 mL/min/1.73 m² (interquartile range, $1.4-4.8 \text{ mL/min}/1.73 \text{ m}^2$). The mean predialysis serum potassium level was 5.0 mmol/L (SD, 0.8 mmol/L). The prevalence of the 6 predialysis serum potassium level categories of $\leq 4.0 \text{ mmol/L}$, $\geq 4.0 \text{ mmol/L}$ to $\leq 4.5 \text{ mmol/L}$, >4.5 mmol/L to \leq 5.0 mmol/L, >5.0 mmol/L to \leq 5.5 mmol/L (reference), >5.5 mmol/L to ≤ 6.0 mmol/L, and >6.0 mmol/L were 10%, 19%, 26%, 22%, 15%, and 8%, respectively. Mean predialysis serum potassium levels in the lowest (SD, \leq 4.0 mmol/L) and highest (SD, >6.0 mmol/L) potassium categories were 3.7 mmol/L (SD, 0.3 mmol/L)

 Table 1. Baseline Characteristics of 1,117 Incident Hemodialysis Patients, Overall and According to 6 Predialysis Serum Potassium

 Level Categories

		Predialysis Serum Potassium Level Category, mmol/L					
Characteristic	All Patients	≤4.0	>4.0 to ≤4.5	>4.5 to ≤5.0	>5.0 to ≤5.5	>5.5 to ≤6.0	>6.0
Total, n (%)	1,117	117 (10)	208 (19)	287 (26)	241 (22)	170 (15)	94 (8)
Age, y	63 (14)	66 (13)	67 (11)	64 (14)	61 (14)	62 (14)	59 (15)
Men, n (%)	649 (58)	70 (60)	117 (56)	164 (57)	147 (61)	94 (55)	57 (61)
Ethnicity, n (%)							
White	1,030 (92)	108 (92)	196 (94)	265 (92)	222 (92)	158 (93)	81 (86)
Black	25 (2)	4 (3)	3 (2)	4 (1)	8 (3)	3 (2)	3 (3)
Asian	60 (6)	5 (4)	9 (4)	18 (6)	11 (5)	9 (5)	8 (9)
Other	2 (0)	0 (0)	0 (0)	0 (0)	0 (0)	0 (0)	2 (2)
BMI, kg/m²	24.7 (4.4)	24.5 (4.4)	25.0 (4.8)	25.2 (4.5)	24.1 (3.9)	24.8 (4.3)	24.1 (4.2)
SGA, n (%)							
1-3; severe PEW	70 (7)	17 (15)	17 (9)	14 (5)	8 (4)	9 (6)	5 (6)
4-5; moderate PEW	275 (26)	31 (28)	65 (34)	63 (23)	54 (24)	40 (25)	22 (27)
6-7; normal PEW	690 (67)	62 (57)	109 (57)	193 (72)	159 (72)	112 (70)	55 (67)
Current smoking, n (%)	288 (26)	26 (22)	54 (26)	68 (24)	74 (31)	38 (22)	28 (30)
Primary kidney disease, n (%)							
Diabetic nephropathy	176 (16)	18 (16)	35 (17)	42 (15)	37 (15)	30 (18)	14 (15)
Glomerulonephritis	113 (10)	10 (9)	12 (6)	30 (11)	24 (10)	21 (13)	16 (17)
Renal vascular disease	245 (22)	24 (21)	55 (27)	62 (22)	52 (22)	30 (18)	22 (24)
Other	574 (52)	64 (55)	104 (51)	153 (53)	128 (53)	85 (51)	40 (44)
Kt/V,ª wk	3.5 (2.8-4.0)	3.7 (3.2-4.3)	3.6 (2.8-4.1)	3.4 (2.8-4.0)	3.3 (2.7-3.8)	3.3 (2.7-3.9)	3.3 (2.7-3.9)
Residual eGFR, ^{a,b} mL/min/ 1.73 m ²	3.5 (1.4-4.8)	3.5 (2.0-6.3)	3.8 (2.0-5.7)	3.5 (1.6-5.1)	3.0 (1.6-4.1)	2.3 (1.0-3.8)	1.5 (0.5-3.4)
Dialysis frequency, n (%)							
1 d/wk	10 (1)	1 (1)	2 (1)	6 (2)	0 (0)	1 (1)	0 (0)
2 d/wk	430 (39)	44 (38)	81 (39)	120 (42)	94 (39)	56 (34)	35 (37)
3 d/wk	671 (60)	72 (62)	122 (59)	161 (56)	147 (61)	110 (66)	59 (63)
Systolic BP,° mm Hg	148 (19)	144 (21)	146 (19)	150 (18)	148 (20)	148 (20)	149 (17)
Diastolic BP,° mm Hg	80 (10)	79 (10)	78 (10)	80 (9)	81 (10)	80 (11)	82 (10)
History of diabetes, n (%)	260 (24)	31 (27)	55 (27)	66 (23)	51 (21)	37 (22)	20 (22)
History of cardiovascular disease, ^d n (%)	354 (32)	44 (38)	79 (38)	92 (32)	70 (29)	50 (29)	19 (20)
History of chronic lung disease, n (%)	98 (9)	11 (10)	23 (11)	22 (8)	21 (9)	13 (8)	8 (9)
History of malignancy, n (%)	131 (12)	13 (11)	29 (14)	31 (11)	26 (11)	24 (15)	8 (9)
Antihypertensives, n (%)	884 (80)	93 (80)	160 (77)	233 (82)	195 (81)	133 (79)	70 (78)
Insulin, n (%)	149 (13)	18 (16)	33 (16)	31 (11)	31 (13)	22 (13)	14 (15)
Lipid-lowering drugs, n (%)	284 (26)	30 (26)	62 (30)	77 (27)	59 (25)	36 (22)	20 (22)
Potassium, mmol/L	5.0 (0.8)	3.7 (0.3)	4.3 (0.1)	4.8 (0.1)	5.3 (0.1)	5.7 (0.1)	6.5 (0.3)
Hemoglobin, mmol/L	6.7 (0.9)	6.6 (1.1)	6.6 (1.0)	6.7 (0.8)	6.7 (0.8)	6.7 (0.9)	6.5 (0.8)
Bicarbonate, mmol/L	22 (3)	24 (3)	23 (3)	22 (3)	22 (3)	21 (4)	21 (3)
Cholesterol, mmol/L	4.7 (1.2)	4.8 (1.4)	4.9 (1.3)	4.6 (1.2)	4.6 (1.1)	4.9 (1.3)	4.7 (1.0)
Phosphate, mmol/L	1.9 (0.6)	1.6 (0.6)	1.7 (0.5)	1.8 (0.5)	1.9 (0.5)	2.0 (0.6)	2.2 (0.7)
Albumin, g/L	35 (6)	35 (6)	36 (6)	36 (6)	35 (6)	36 (6)	36 (6)
nPCR, ^e g/kg/d	1.0 (0.2)	1.0 (0.2)	1.0 (0.3)	1.0 (0.2)	1.0 (0.2)	1.0 (0.2)	1.0 (0.2)

Note: Continuous data are expressed as the mean (± standard deviation), unless otherwise indicated. Measurements were conducted prior to dialysis. Abbreviations: BMI, body mass index; BP, blood pressure; eGFR, estimated glomerular filtration rate; Kt/V, *K*- clearance of urea, *t*- time, *V*- volume of distribution of urea, approximately equal to patient's total body water; nPCR, normalized protein catabolic rate; PEW, protein-energy wasting; SGA, Subjective Global Assessment. ^aMedian (interquartile range).

^bThe residual eGFR was based on combined urea and creatinine clearance and corrected for body surface.

^cMean systolic and diastolic BP values shown are prior to dialysis, as measured over the previous 2 weeks.

^dCardiovascular disease was defined as any history of a cerebral vascular accident, a myocardial infarction peripheral vascular disease, or heart failure.

^eThe nPCR was calculated using the Watson nomogram.

and 6.5 mmol/L (SD, 0.3 mmol/L), respectively. Table 1 presents the baseline characteristics for all patients and according to the 6 predialysis serum potassium level categories.

Compared with the reference category (>5.0 to \leq 5.5 mmol/L), patients in the lowest potassium level category were older, smoked less often, and had lower SGA scores, whereas

	Time-Dependent Predialysis Serum Potassium Level Category, mmol/L						
All-Cause Mortality	≤4.0	>4.0 to ≤4.5	>4.5 to ≤5.0	>5.0 to ≤5.5	>5.5 to ≤6.0	>6.0	
Person-years	199.5	451.7	722.1	712.2	506.2	323.3	
Deaths from all causes	52	86	154	122	78	63	
Deaths from all causes/ 100 person-years	26.1 (19.0-33.1)	19.0 (15.0-23.1)	21.3 (17.9-24.7)	17.1 (14.1-20.2)	15.4 (12.0-18.8)	19.5 (14.7-24.3)	

 Table 2.
 Absolute All-Cause Mortality Rates (95% Confidence Intervals) According to 6 Time-Dependent Predialysis Serum

 Potassium Level Categories During 10-Year Follow-up of 1,117 Incident Hemodialysis Patients

those in the highest predialysis serum potassium level category were younger and had lower residual kidney functions. Phosphate levels increased with higher potassium level categories, whereas bicarbonate levels decreased.

Mortality Risk

During 10 years of follow-up, 75 (7%) patients receiving hemodialysis switched to peritoneal dialysis and were thus censored. Median survival for all patients during 10 years of follow-up was 3.93 years. In total, 555 deaths were observed over 2,915 person-years, resulting in an overall crude all-cause mortality rate of 19.0 per 100 patient-years (95% CI, 17.5-20.7 per 100 patient-years). Approximately 29% of all deaths were due to cardiac causes. The Kaplan-Meier curves showing the survival probabilities for each baseline predialysis serum potassium level category during 10 years of follow-up are demonstrated in Figure S1. Table 2 shows the mortality rates (95% CIs) per 100 patient-years during 10 years of follow-up according to the 6 time-dependent potassium level categories. The absolute risk of death was clearly increased in the lowest predialysis serum potassium level category of ≤4.0 mmol/ L compared with >5.0 to ≤ 5.5 mmol/L, which corresponded to an excess rate of approximately 9 deaths/100 patient-years. The highest predialysis serum potassium level category of >6.0 mmol/L corresponded to an excess rate of approximately 2 deaths/100 patient-years compared with the predialysis serum potassium level category of >5.0 to \leq 5.5 mmol/L.

After checking the proportional hazards assumption, we found no sign of violation. Table 3 shows the crude and

multivariable adjusted relationship between the 6 timedependent predialysis serum potassium level categories and 6-month mortality during 10 years of follow-up. Additional adjustments for residual kidney function or SGA did not materially attenuate the relationship between predialysis serum potassium level and mortality. The HR of time-dependent predialysis serum potassium \leq 4.0 mmol/L was 1.42 (95% CI, 1.01-1.99), implying that hypokalemia is a 1.4-fold stronger risk factor compared with predialysis serum potassium levels of >5.0 and \leq 5.5 mmol/L, whereas predialysis serum potassium levels of >6.0 mmol/L resulted in an HR of 1.32 (95% CI, 0.97-1.81).

U-Shaped Relationship

Figure 1 shows the U-shaped relationship between timedependent predialysis serum potassium level and 6month mortality during 10 years of follow-up, expressed using the multivariable adjusted HR, with a nadir at approximately 5.1 mmol/L. HRs for death increased substantially below a predialysis serum potassium level of ≤ 4.5 mmol/L and above >5.7 mmol/L, with the effects of lower predialysis serum potassium levels being more pronounced. For example, patients receiving hemodialysis with a predialysis serum potassium level of 4.0 mmol/L compared with the optimum level of 5.1 mmol/L had an almost 1.4-fold increased risk of death.

Sensitivity Analyses

We performed 4 sensitivity analyses. First, additional adjustments for the nutritional markers phosphate, bicarbonate, albumin, and normalized protein catabolic rate did

Time-Dependent Predialysis Serum Potassium Level Category, mmol/L	Hazard Ratio (95% CI)							
	Crude	Age and Sex Adjusted	Model 1	Model 2	Model 3			
≤4.0	1.64 (1.18-2.29)	1.47 (1.05-2.05)	1.40 (1.00-1.96)	1.48 (1.05-2.07)	1.42 (1.01-1.99)			
>4.0 to ≤4.5	1.22 (0.92-1.62)	1.13 (0.85-1.50)	1.04 (0.78-1.38)	1.10 (0.82-1.46)	1.09 (0.82-1.45)			
>4.5 to ≤5.0	1.28 (1.00-1.63)	1.18 (0.92-1.51)	1.17 (0.91-1.50)	1.22 (0.95-1.57)	1.21 (0.94-1.56)			
>5.0 to ≤5.5	1	1	1	1	1			
>5.5 to ≤6.0	0.94 (0.70-1.26)	0.96 (0.72-1.29)	0.98 (0.73-1.31)	0.96 (0.71-1.29)	0.95 (0.71-1.28)			
>6.0	1.23 (0.90-1.69)	1.39 (1.02-1.90)	1.38 (1.01-1.89)	1.36 (0.99-1.85)	1.32 (0.97-1.81)			

 Table 3. Hazard Ratios With 95% Confidence Intervals of 6-Month All-Cause Mortality According to the 6 Categories of Time

 Dependent Predialysis Serum Potassium Levels in 1,117 Incident Hemodialysis Patients During 10 Years of Follow-up

Note: The serum potassium level of 5.0 to 5.5 mmol/L was taken as the reference category. Model 1 was adjusted for age, sex, current smoking, history of diabetes mellitus, and history of cardiovascular disease. Model 2 had an additional adjustment for residual kidney function. Model 3 (full model) had an additional adjustment for residual kidney function and subjective global assessment score. Abbreviation: Cl. confidence interval.



Figure 1. Hazard ratio for 6-month all-cause mortality with 95% confidence intervals (dotted lines) related to time-dependent predialysis serum potassium levels in 1,117 incident hemodialysis patients during 10 years of follow-up, calculated from our 4-knot restricted cubic spline and adjusted for age, sex, current smoking, a history of diabetes mellitus, and a history of cardiovascular disease.

not materially alter the results of our main analysis (Table S1). Second, considering cardiac death as an outcome attenuated the results due to the loss of power but showed a similar direction of the relationship (Table S2). Third, repeating our analyses without censoring for patients who switched to peritoneal dialysis also did not substantially change the effect between predialysis serum potassium level and death (Table S3). Finally, considering predialysis serum potassium level as a fixed category at baseline did attenuate the strength of the relationship between predialysis serum potassium level category and 10-year all-cause mortality as expected due to the dilution of the effect of potassium over time (Tables S4 and S5).

DISCUSSION

This prospective study among exclusively incident hemodialysis patients showed a U-shaped relationship between predialysis serum potassium level and death during 10 years of follow-up, with an optimum level of approximately 5.1 mmol/L. Compared with this optimum level, low ($\leq 4 \text{ mmol/L}$) and high (>6 mmol/L) predialysis serum potassium concentrations were 1.4- and 1.3-fold stronger risk factors for death after multivariable adjustment, respectively.

Our results are in line with the study by Torlén et al,³ including >1,11,000 prevalent hemodialysis patients showing adjusted HRs for all-cause mortality for time-dependent serum potassium levels of <3.5 mmol/L and \geq 5.5 mmol/L of 2.0 (95% CI, 1.8-2.1) and 1.2 (95%

CI, 1.2-1.3) compared with the reference category of 4.0-4.5 mmol/L, respectively. A limitation of this previous study is the inclusion of prevalent patients, which may have resulted in selection bias. In addition, estimates were not adjusted for the potential confounders residual kidney function and nutritional status. In contrast, another study including 55,000 prevalent hemodialysis patients showed low adjusted HRs for death of 1.0 (95% CI, 1.0-1.1) and 1.1 (95% CI, 1.0-1.2) for serum potassium <4.0 mmol/L and >6.0 mmol/L, respectively, compared to the reference category of 4.0 to 5.0 mmol/L.12 Next to selection bias and their relatively low reference category, another possible explanation for their low HRs could be the use of serum potassium as a fixed variable in their model over a median follow-up of 16.5 months. When considering predialysis serum potassium level as a fixed value at baseline, we had a similar finding, namely, a weakening of the observed effect over time most likely due to dilution.

The relatively high optimum predialysis serum potassium level of 5.1 mmol/L that we found is consistent with 2 previous studies. Kovesdy et al¹⁹ reported that predialysis serum potassium levels between 4.6 and 5.3 mmol/L were associated with the highest 3-year survival rate in >81,000 prevalent hemodialysis patients. Pun et al²⁰ found that a predialysis serum potassium level of 5.1 mmol/L resulted in the lowest risk of sudden cardiac arrest.

Limitations of these 2 previous studies were again the inclusion of prevalent patients and, in the study by Pun et al,²⁰ failure to adjust for potential confounders such as smoking, residual kidney function, and nutritional status.

Serum potassium concentration rapidly decreases by approximately 1 mEq/L in the first hour of hemodialysis treatment, when the blood-to-dialysate gradient is greatest, and an additional lowering of 1 mEq/L occurs over the next 2 hours. A rapid rebound of serum potassium occurs because of efflux from the intracellular compartment after completion of the dialysis session.⁷ Patients with a relatively low predialysis serum potassium level may experience more severe or prolonged hypokalemia after the session, which may explain the increased risk of death that we found.²¹ Another explanation could be that low predialysis serum potassium level is a proxy for malnourishment, which is a strong risk factor for death.^{22,23} However, adjusting for SGA score, the normalized protein catabolic rate, serum phosphate, bicarbonate, and albumin, as indicators of malnutrition, did not materially attenuate the relationship between predialysis serum potassium level and death. Finally, low total body potassium is also a risk factor for death.⁸ Hemodialysis treatment may reduce total body potassium by depleting intracellular potassium stores. Although serum potassium level does not necessarily reflect the intracellular potassium, a low predialysis concentration may serve as a proxy of a relatively low total body potassium level.

There are several strengths to our study. First, to our knowledge, this is the first study that included only incident hemodialysis patients. All previous studies

investigating the relationship between predialysis serum potassium level and death mainly included prevalent hemodialysis patients, thus being susceptible to survivor bias.^{3,4,11,12,19,23,24} This is a form of selection bias that occurs when the risk of an outcome is estimated from data collected at a given time point among survivors rather than on data gathered in a group of incident cases.¹⁰ As with other biases, an increased study size cannot compensate for survivor bias.²⁵ Second, all measurements were performed according to the study protocol, and, therefore, information bias is unlikely. This is in contrast to the previous studies in which data were collected from clinical records, potentially resulting in information bias as data were collected for a clinical reason. Third, we adjusted for smoking, an important confounder that was unavailable in the majority of the previous studies and, therefore, not included in the model.^{4,11,12,23,24} Reverse causation owing to inadequate control for smoking status can distort the true relationship between hypokalemia and the risk of death because smoking is associated with both decreased serum potassium level and an increased risk of death.²⁶ Finally, by modeling potassium freely in our restricted cubic spline analysis, we could establish the optimum predialysis serum potassium level as a reference category and incorporate it into our main time-dependent analysis. As this reference was found to be higher than those used in previous studies, this may have allowed for a more valid estimation of the HRs associated with dyskalemia, particularly hypokalemia. Using a lower reference category, as most previous studies did, could have resulted in the underestimation of the relative relationship between hypokalemia and mortality.

Nevertheless, our study has some limitations. First, as with most studies, we encountered missing data. To maintain power and minimize any bias, we used multiple imputations to account for these missing data. Second, even though we updated predialysis serum potassium level as a time-dependent variable, we could only do so for every 6 months. As the serum predialysis potassium level fluctuates, shorter intervals between measurements would have allowed for more precise estimations of the mortality risk and less "dilution" of the effect. Third, as we did not have information on peridialytic changes in serum potassium levels, the serum-to-dialysate potassium gradient, or postdialysis serum potassium levels, we could not consider the effects of these factors on the mortality risk. Fourth, we used all-cause mortality as the primary outcome, which is unequivocal, whereas the secondary outcome, cardiac death, can be nondifferentially misclassified. Considering cardiac death as an outcome showed slightly weaker but similar results. In general, nondifferential misclassification results in an underestimation of the effect.²⁵

In conclusion, we found a U-shaped relationship between predialysis serum potassium levels and 6-month allcause mortality in incident hemodialysis patients in the first 10 years of follow-up. Our results indicate an optimum predialysis serum potassium level of approximately 5.1 mmol/L. Low and high predialysis serum potassium levels resulted in 1.4- and 1.3-fold stronger risk factors for death, respectively, compared with the optimum level. If proven causal, the clinical implication of these results is that potassium-lowering therapy should be used with caution in patients receiving hemodialysis with normal or low serum potassium levels before the dialysis session. Furthermore, as low predialysis serum potassium level could result from malnourishment, the associated mortality risk emphasizes the importance of preventing nutritional disorders in patients receiving hemodialysis.

SUPPLEMENTARY MATERIAL

Supplementary File (PDF)

Figure S1: Kaplan-Meier curves showing the survival probability for each baseline serum potassium level category over 10 years of follow-up.

Table S1: Hazard ratios with 95% confidence intervals of 6-month all-cause mortality according to the 6 categories of time-dependent predialysis serum potassium levels in 1,117 incident hemodialysis patients during 10 years of follow-up, with additional adjustments for nutritional markers.

Table S2: Hazard ratios with 95% confidence intervals of 6-month cardiac death according to the 6 categories of time-dependent predialysis serum potassium levels in 1,117 incident hemodialysis patients during 10 years of follow-up.

Table S3: Hazard ratios with 95% confidence intervals of 6-month all-cause mortality according to the 6 categories of time-dependent predialysis serum potassium levels in 1,117 incident hemodialysis patients over 10 years of follow-up, without censoring at the time of a switch to peritoneal dialysis.

Table S4: Hazard ratios with 95% confidence intervals of 10-year allcause mortality according to the 6 categories of predialysis serum potassium levels fixed at baseline in 1,117 incident hemodialysis patients.

Table S5: Hazard ratios with 95% confidence intervals of 10-year cardiac death according to the 6 categories of predialysis serum potassium levels fixed at baseline in 1,117 incident hemodialysis patients.

ARTICLE INFORMATION

Netherlands Cooperative Study on the Adequacy of Dialysis-2 (NECOSAD) Study Investigators: A.J. Apperloo, J.A. Bijlsma, M. Boekhout, W.H. Boer, P.J.M. van der Boog, H.R. Büller, M. van Buren, F.Th. de Charro, C.J. Doorenbos, M.A. van den Dorpel, A. van Es, W.J. Fagel, G.W. Feith, C.W.H. de Fijter, L.A.M. Frenken, W. Grave, J.A.C.A. van Geelen, P.G.G. Gerlag, J.P.M.C. Gorgels, R.M. Huisman, K.J. Jager, K. Jie, W.A.H. Koning-Mulder, M.I. Koolen, T.K. Kremer Hovinga, A.T.J. Lavrijssen, A.J. Luik, J. van der Meulen, K.J. Parlevliet, M.H.M. Raasveld, F.M. van der Sande, M.J.M. Schonck, M.M.J. Schuurmans, C.E.H. Siegert, C.A. Stegeman, P. Stevens, J.G.P. Thijssen, R.M. Valentijn, G.H. Vastenburg, C.A. Verburgh, H.H. Vincent, and P.F. Vos.

Authors' Full Names and Academic Degrees: Esther N.M. de Rooij, MD, Friedo W. Dekker, PhD, Saskia Le Cessie, PhD, Ewout J. Hoorn, MD, PhD, Johan W. de Fijter, MD, PhD, and Ellen K. Hoogeveen, MD, PhD

Authors' Affiliations: Department of Clinical Epidemiology (ENMdR, FWD, SLC, EKH), and Department of Nephrology (ENMdR, JWdF,

EKH), Leiden University Medical Center, Leiden, The Netherlands; Department of Internal Medicine, Division of Nephrology and Transplantation, Erasmus Medical Center, Rotterdam, The Netherlands (EJH); and Department of Nephrology, Jeroen Bosch Hospital, Den Bosch, The Netherlands (EKH)

Address for Correspondence: Esther N.M. de Rooij, MD, Department of Nephrology, Leiden University Medical Center, Building 1, C7-Q, Albinusdreef 2, 2333 ZA, Leiden, The Netherlands. Email: e.n.m.rooij@lumc.nl

Authors' Contributions: Research idea and study design: ENMdR, EKH, FWD, JWdF; data acquisition: FWD; data analysis/ interpretation: ENMdR, EKH, FWD, JWdF; statistical analysis: ENMdR, EKH, SLC, FWD; supervision or mentorship: EKH, FWD, JWdF, SLC, EJH. Each author contributed important intellectual content during manuscript drafting or revision and accepts accountability for the overall work by ensuring that questions pertaining to the accuracy or integrity of any portion of the work are appropriately investigated and resolved.

Support: The NECOSAD was supported by grants from the Dutch Kidney Foundation and the Dutch National Health Insurance Board.

Financial Disclosure: The authors declare that they have no relevant financial interests.

Acknowledgements: The authors thank the investigators and study nurses of the participating dialysis centers and the data managers of the Netherlands Cooperative Study on the Adequacy of Dialysis (NECOSAD) study for collection and management of data. The authors thank the nursing staff of the participating dialysis centers and the staff of the NECOSAD trial office for their invaluable assistance in the collection and management of data for this study.

Peer Review: Received June 9, 2021. Evaluated by 2 external peer reviewers, with direct editorial input by a Statistical Editor and the Editor-in-Chief. Accepted in revised form August 18, 2021.

REFERENCES

- 1. Choi HY, Ha SK. Potassium balances in maintenance hemodialysis. *Electrolyte Blood Press*. 2013;11(1):9-16.
- Palmer BF, Clegg DJ. Physiology and pathophysiology of potassium homeostasis. Adv Physiol Educ. 2016;40(4):480-490.
- Torlén K, Kalantar-Zadeh K, Molnar MZ, Vashistha T, Mehrotra R. Serum potassium and cause-specific mortality in a large peritoneal dialysis cohort. *Clin J Am Soc Nephrol.* 2012;7(8):1272-1284.
- Brunelli SM, Du Mond C, Oestreicher N, Rakov V, Spiegel DM. Serum potassium and short-term clinical outcomes among hemodialysis patients: impact of the long interdialytic interval. *Am J Kidney Dis.* 2017;70(1):21-29.
- Wiegand CF, Davin TD, Raij L, Kjellstrand CM. Severe hypokalemia induced by hemodialysis. *Arch Intern Med.* 1981;141(2):167-170.
- 6. Rossignol P, Lamiral Z, Frimat L, et al. Hyperkalaemia prevalence, recurrence and management in chronic haemodialysis: a prospective multicentre French regional registry 2-year survey. *Nephrol Dial Transplant*. 2017;32(12):2112-2118.
- 7. Wouda RD, Vogt L, Hoorn EJ. Personalizing potassium management in patients on haemodialysis. *Nephrol Dial Transplant*. 2021;36(1):13-18.
- Dolson GM, Ellis KJ, Bernardo MV, Prakash R, Adrogué HJ. Acute decreases in serum potassium augment blood pressure. *Am J Kidney Dis.* 1995;26(2):321-326.
- Kovesdy CP, Matsushita K, Sang Y, et al. Serum potassium and adverse outcomes across the range of kidney function: a CKD

Prognosis Consortium meta-analysis. *Eur Heart J.* 2018;39(17):1535-1542.

- Jager KJ, Stel VS, Zoccali C, Wanner C, Dekker FW. The issue of studying the effect of interventions in renal replacement therapy—to what extent may we be deceived by selection and competing risk? *Nephrol Dial Transplant*. 2010;25(12):3836-3839.
- Yusuf AA, Hu Y, Singh B, Menoyo JA, Wetmore JB. Serum potassium levels and mortality in hemodialysis patients: a retrospective cohort study. *Am J Nephrol.* 2016;44(3):179-186.
- 12. Karaboyas A, Zee J, Brunelli SM, et al. Dialysate potassium, serum potassium, mortality, and arrhythmia events in hemodialysis: results from the Dialysis Outcomes and Practice Patterns Study (DOPPS). *Am J Kidney Dis.* 2017;69(2): 266-277.
- **13.** de Mutsert R, Snijder MB, van der Sman-de Beer F, et al. Association between body mass index and mortality is similar in the hemodialysis population and the general population at high age and equal duration of follow-up. *J Am Soc Nephrol.* 2007;18(3):967-974.
- Visser R, Dekker FW, Boeschoten EW, Stevens P, Krediet RT. Reliability of the 7-point subjective global assessment scale in assessing nutritional status of dialysis patients. *Adv Perit Dial*. 1999;15:222-225.
- Rastegar A. Serum potassium. In: Walker HK, Hall WD, Hurst JW, eds. *Clinical Methods: The History, Physical, and Laboratory Examinations.* 3rd ed. Butterworths; 1990.
- Mansournia MA, Etminan M, Danaei G, Kaufman JS, Collins G. Handling time varying confounding in observational research. *BMJ*. 2017;359:j4587.
- Harrell FE Jr, Lee KL, Califf RM, Pryor DB, Rosati RA. Regression modelling strategies for improved prognostic prediction. *Stat Med.* 1984;3(2):143-152.
- Moons KG, Donders RA, Stijnen T, Harrell FE Jr. Using the outcome for imputation of missing predictor values was preferred. J Clin Epidemiol. 2006;59(10):1092-1101.
- 19. Kovesdy CP, Regidor DL, Mehrotra R, et al. Serum and dialysate potassium concentrations and survival in hemodialysis patients. *Clin J Am Soc Nephrol.* 2007;2(5):999-1007.
- Pun PH, Lehrich RW, Honeycutt EF, Herzog CA, Middleton JP. Modifiable risk factors associated with sudden cardiac arrest within hemodialysis clinics. *Kidney Int.* 2011;79(2):218-227.
- Ohnishi T, Kimachi M, Fukuma S, Akizawa T, Fukuhara S. Postdialysis hypokalemia and all-cause mortality in patients undergoing maintenance hemodialysis. *Clin J Am Soc Nephrol.* 2019;14(6):873-881.
- Kang SS, Chang JW, Park Y. Nutritional status predicts 10-year mortality in patients with end-stage renal disease on hemodialysis. *Nutrients*. 2017;9(4):399.
- 23. Hwang JC, Wang CT, Chen CA, Chen HC. Hypokalemia is associated with increased mortality rate in chronic hemodialysis patients. *Blood Purif.* 2011;32(4):254-261.
- 24. Lee S, Kang E, Yoo KD, et al. Lower serum potassium associated with increased mortality in dialysis patients: a nationwide prospective observational cohort study in Korea. *PLoS One*. 2017;12(3):e0171842.
- 25. Rothman KJ. *Epidemiology: An Introduction*. Oxford University Press; 2002.
- 26. Wannamethee SG, Lever AF, Shaper AG, Whincup PH. Serum potassium, cigarette smoking, and mortality in middle-aged men. *Am J Epidemiol.* 1997;145(7):598-606.

